SUCCESSION PLANNING IN SMALL AND MEDIUM ENTERPRISES IN THE TAKORADI METROPOLIS OF GHANA.

By
MENSAH, JUDITH
(PG3072509)

A Thesis submitted to the Institute of Distance Learning, Kwame Nkrumah University of Science and Technology in partial fulfillment of the requirements for the degree of

COMMONWEALTH EXECUTIVE MASTERS OF BUSINESS ADMINISTRATION

SEPTEMBER 2012

DECLARATION

I hereby declare that this submission is my own work towards the executive masters of Business Administration and that, to the best of my knowledge, it contains no material previously published by another person nor material which has been accepted for the award of any other degree of the University, except where due acknowledgement has been made in the text.

MENSAH JUDITH, PG3072509 Student Name & ID	Signature	Date		
Certified by:				
Dr. Michael Adusei Boadu Supervisor Name	Signature	Date		
Certified By:				
Prof. I.K. Dontwi Dean, IDL	Signature	Date		

DEDICATION

In remembrance of my grandmother, Mrs Margaret Ellis-Mensah.



ACKNOWLEDGEMENTS

I got a lot of support during the writing process of this dissertation and want to address some grateful thanks to certain persons.

Firstly, I want to thank my supervisor, Dr M.A Boadu for his guidance and valuable feedback he provided me. I want to express special thanks to the respondents who spent their scarce time and shared their valuable knowledge and experience with me. To Michael Bosomtwe of Radford University College, a valuable friend for his support throughout the dissertation write up. I say bravo.

Last but not the least to my priceless family for their support and advice especially my husband Mr. Samuel Emmanful-Arkah.



ABSTRACT

The objective of the study was to find out factors affecting succession planning in Small and Medium Enterprises (SMEs) in Takoradi Metropolis and again to suggest ways of enhancing succession planning in Small and Medium Enterprises in Takoradi. Descriptive survey was used for the study. Purposive sampling was used in selecting 125 SMEs for the study. A respondent each was required from the 125 SMEs and so stratified sampling was used to group the respondents into owners and workers. 60% quota was given to owners and 40% to workers through quota sampling. Convenience sampling was then used to sample the required number. Data was gathered through the use of questionnaire and analysed using percentages and frequency tables. The major findings of the study are that lack of trust was one of the factors affecting succession planning in SMEs. This was because many SMEs do not have clear objectives and policies of succession planning for the enterprise. The study found out that succession was not based on merit or succession was based on the relationship with the founder of the business. The study revealed that good management practices and clear objectives of the SMEs in terms of who would take over or assume position when the owner or founder retires or dies could relate to succession planning. The study again identified that many SMes do not involve their family or relatives in the operations of the business. The study recommended that SMEs should train prospective successors for the enterprises. Again there should be succession laws to regulate the smooth transition to new administration. Finally the study recommends that successors for the business should be based on merit to ensure that right and qualified people manage the enterprise.

TABLE OF CONTENTS

TITLE			PAGE
DECLARATION			::
DEDICATION			
ACKNOWLEDGEMENT		iv	
ABTRACT			V
TABLE CONTENTS	KNII	ICT	OFvi
LIST	OF	V	TABLES
CHAPTER	W	May .	ONE
INTRODUCTION		•••••	1
1.1 BACKGROUND	ТО	THE	STUDY
1.2 STATEMENT PROBLEM		OF4	THE
1.3 OBJECTIVES	OF	THE5	STUDY
1.4 RESEARCH		6	QUESTIONS
1.5 SIGNIFICANCE	OF	THE6	STUDY
1.6 DELIMITATION	OF	THE7	STUDY
1.7 LIMITATION	OF	THE7	STUDY
1.8 DEFINITION		OF	TERMS
1.9 ORGANISATION	OF	THE	STUDY
CHAPTER			TWO

REVIEW	OF	RELAT	ΓED	LITERATURE
2.1				INTRODUCTION
2.2 CONCEPT C	OF SMALL AN			ENTERPRISES (SME)
10 2.3		CESSION	4.0	PLANNING
2.4				CONCLUSION33
CHAPTER	K	NU	ST	THREE34
METHODOLOGY	Y 		•••••	34
3.1				INTRODUCTION
3.2		ESEARCH		34 DESIGN
3.3				POPULATION
3.4 PROCEDURE	SAMPLING		AND 35	SAMPLING
3.5	SOURCE		OD	DATA 36
3.6 INSTRUMENTS 3.7		PILOT		RESEARCH 36 TESTING
3.8		COLLEC	ΓΙΟΝ	PROCEDURE
3.9	To I	DATA		ANALYSIS
		SANE NO		38
CHAPTER			•••••	FOUR 39
RESULTS		AND		DISCUSSIONS
4.1				INTRODUCTION39
4.2 DEMOGRAPH 39	IC DATA OF RES	SPONDENTS.		

4.3 EDUCATION A	AL QUALIF	ICATIONS O	F RESPON	DENTS		
4.4 NUMBER O40	F YEARS	RESPONDE	NTS HAS	WORKED	IN THE	SMES
4.5 RESEARCH	IN S	N 1: WHA' ME IN	TAK			
4.6 RESEARCH ENHANCE	QUESTION SUCCESS	2: WHAT	FACTORS PLANNIN			
CHAPTER		1.7.5.1.			51	FIVE
SUMMARY RECOMMENDA	OF	FINDING				AND
5.1 FINDINGS		SUMM		5	1	OF
5.2CONCLUSION5.3		M				52
RECOMMENDAT	ION					52
5.4	SUGGES'	ΓΙΟΝ	F	OR	= FI	JTURE
STUDY			53			
REFERENCES					•••••	•••••
58						
APPENDIX		•••••	•••••	• • • • • • • • • • • • • • • • • • • •	• • • • • • • • • • • • • • • • • • • •	•••••
59		SANE				

LIST OF TABLES

Table 4.1 The distribution of respondents educational by qualification.....40 respondents The number Table 4.2 of years have worked in the SMEs.....41 affecting planning Table 4.3 Factors succession in SMEs.....41 Table 4.4 Strategies that can be adopted to enhance succession planning in SMEs.....48



CHAPTER ONE

INTRODUCTION

1.1.BACKGROUND TO THE STUDY

Small and medium enterprises (SMEs) have been considered as the engine of economic growth and for promoting equitable development. The major advantage of the sector is its employment potential at low capital cost. The labour intensity of the SME sector is much higher than that of the large enterprises. The sector is a nursery of entrepreneurship, often driven by individual creativity and innovation. It is noted that the SME sector is the main driving force behind job creation, poverty reduction, wealth creation, income distribution and reduction in income disparities. SME sector forms the vanguard of the modern enterprise sector and presents the propelling force of economic modernization and growth in the country (Aremu, 2004).

Jasani (2002) postulates that owner-managed businesses employ half the world's workforce and generate more than half the world's Gross Domestic Product(GDP). However, the needs, problems and prospects of SMEs are often ignored. According to Jasani, there is the need to change in management thinking and technological innovations if small and medium enterprises are to enhance their competitive edge. Jasani suggests that while the medium and large-scale enterprises seemed to be aligning themselves to prepare for the onset of globalisation and

technology advancement, the small-scale enterprises appeared to be lagging behind in these important areas.

Owner-managed and family businesses tend to be driven by the qualities and skills of their principals which is their great strength. They create challenging problems, as the aspirations of the owners may not always be in harmony with the needs of the business. Jasani reports in his study that majority of Malaysian family run businesses that are small scale enterprises employing below 52 staff, and are still very the founder with much managed by activities focused manufacturing, retailing or construction. It is interesting to note that people having six or more years of work experience, indicating that family businesses were commenced by people with appropriate experience, started most of these businesses. Further, Jasani identifies three key areas of concerns for business owners and these include succession, funding and professional management. He asserts that irrespective of the size of the business, most people feel that there should only be one management successor chosen from the family or the management. There is also a general concern about losing control if the business owners were to involve outsiders to finance growth, especially among the small scale enterprises. According to Amajit (2004), many people do not understand that small scale and family businesses are social organisations and need to be treated as such. Amajit posits that in India a feudal set up with succession planning by the family patriarch being done too late in the life span of the business when such planning should actually happen in the first generation itself. To him,

businesses are meant to create wealth and this wealth should be shared by all in the family.

In recent times, CAL Bank has engaged small and medium scale businesses in Kumasi on the need for a succession plan to sustain their operations. For many small business owners, maintaining positive cash flow and a stable balance sheet can be an ongoing battle that consumes virtually all of their time. Retirement therefore seems not be part of their business plan, let alone plans to hand over the business. Studies suggest that establishing a sound business succession plan is beneficial for business owners. The Bank observed that lack of trust is a major factor influencing SME owners not to institute succession plans. Statistics show that about 70 percent of SME businesses collapse because of lack of succession planning (Domfeh, 2011).

The relative importance of small and medium scale enterprise in advanced and developing countries has led and would continue to lead to a reconsideration of the role of Small and Medium Scale Enterprises in the economy of nations. Within the European Union, in 1998, out of 19 million enterprises, SMEs accounted for 99.8 percent and two thirds of the jobs in the region were created by SMEs. The strong performance of the US economy in recent years has been driven largely by the creation of SMEs (Kyeremanteng, 2007). It is estimated that SMEs employ 22% of the adult population in the developing countries (Kayanula & Quartey, 2000). The role play by SME businesses in the Ghanaian economy cannot be over-emphasized. For example, 70% of all

industrial establishments are in the Small and Medium Enterprises Sector as well as 85% of manufacturing employment comes from the Sector. The sector employs about 15.5% of the labour force in Ghana and it contributes 6% of the GDP (Kayanula & Quartey, 2000). Thus, the development of many countries is measured by such indices as the level of industrialization, modernization, urbanization, gainful and meaningful employment for all those who are able and willing to work, income per capital, equitable distribution of income, and the welfare and quality of life enjoyed by the citizenry. There is no doubt that small scale enterprises exist in most economic environment. Small and Medium Scale Enterprise is an all time necessity. Business transition has posed a problem especially succession in small and medium scale business in Takoradi and part of the country. Hence, there is the need for succession planning in Small and Medium Scale businesses in Takoradi Metropolis. Evidence abound that many SME businesses do not survive the next generation at the demise of the proprietor or founder of the business. It is in the light of this that this research work is undertaken to find out why SME businesses do not survive beyond the founders or proprietors.

1.2.STATEMENT OF THE PROBLEM

The role of the Small and Medium Enterprises (SME) as a catalyst for economic growth and development has been well documented in the economic literature and recognized in most countries. For example, it is estimated that SMEs employ 22% of the adult population in the

developing countries (Kayanula & Quartey, 2000). Kyeremanteng (2007) posits that the SME sector in the Ghanaian economy accounted for 70% of all industrial establishments and 85% of manufacturing employment and to a larger extent overall employment growth in the country comes from the Sector. The sector employs about 15.5% of the labour force in Ghana and it contributes 6% of the GDP. The sector contributes significantly to overall export earnings from the nontraditional exports sector and creates jobs at relatively low capital cost. However, this important sector is bedeviled with succession planning on the retirement of the proprietor or after the demise of the founder. Statistics show that about 70 percent of SME businesses collapse because of lack of succession planning (Domfeh, 2011). Some of the questions bothering the mind of the researcher are what factors are responsible for lack of succession planning in small and medium enterprises and how these challenges can be addressed in SMEs? There is evidence that no research work has been done in the area of succession planning in SME businesses in Takoradi Metropolis. It is in the light of this that the present research is undertaken to find out why SMEs do not survive beyond the founders or proprietors. Hence, there is the need for succession planning in Small and Medium Scale businesses in Takoradi Metropolis.

1.3.OBJECTIVES OF THE STUDY

The general objective of this study was to examine succession planning in small and medium scale enterprises in Takoradi Metropolis. The following specific objectives were proposed to help in achieving the general

objectives of the study:

- To find out the factors affecting succession planning of Small and
 Medium Enterprises in Takoradi Metropolis.
- To suggest ways of enhancing succession planning in Small and Medium Enterprises in Takoradi Metropolis.

1.4. RESEARCH QUESTIONS

Given how essential the succession planning for the survival of SMEs after the death or retirement of the founder, the following research questions are proposed to guide the study:

- 1. What are the factors influencing succession planning in Small and Medium Enterprises in Takoradi Metropolis?
- 2. What can be done to enhance succession planning in Small and Medium Enterprises in Takoradi Metropolis?

1.5. SIGNIFICANCE OF THE STUDY

This study will contribute to knowledge on succession planning in small and medium enterprises in recommending best succession planning for small and medium enterprises. It is hoped that this research will provide a more profound and science-based understanding of the subject at hand and recommendations and solutions to one of the major problems facing the small and medium business community. The findings and recommendations of the research will have a tremendous impact on SMEs in Takoradi Metropolis. Many owners of SMEs are either

not aware of the problem of succession planning or simply avoid talking about it. This makes these businesses vulnerable to succumb when the founder or owner dies.

1.6.DELIMITATION OF THE STUDY

The study was delimited to the SMEs in Takoradi Metropolis of Ghana. Further, the study was delimited to the SMEs workers and owners who were in manufacturing, commerce and service sectors in Takoradi. However, the SMEs which have staff numbering10 or more, as well as workers who have worked for less than two years were not included in the study because they had not gained enough experience on the job.

1.7.LIMITATIONS OF THE STUDY

The questionnaire constructed might not provide all the needed information the researcher might need and this might affect the validity of the results. Respondents were reluctant to provide information because they thought the study would expose the flaws associated with the continuation of the businesses. Further, respondents felt the study would not bring any change in succession planning of the enterprises. This weakness might have affected the results of the study.

Another limitation to the study was the use of non-probability sampling which makes it difficult for the researcher to extrapolate the findings of the study to other population. This might have affected the generalization of the findings to other Small and Medium Enterprises.

1.8. DEFINITION OF TERMS

- 1. Small and Medium Enterprises include a wide range of business activities in agriculture, mining, fishing, industry or manufacturing, construction, retails and whole sales and services in rural and urban settings by servicing local and international markets
- 2. Succession autonomy: the extent of discretionary powers of a successor in relation to running business.
- 3. Communication for succession: the extent of the communication of owner's succession plan to stakeholders.
- 4. Role clarification of outgoing proprietor after succession: the extent of clarification of the position of the proprietor after retirement.
- 5. Succession justice: the extent of justice of succession based on a succession plan agreed by other stakeholders.

1.9. ORGANISATION OF THE STUDY

The study is organised into five chapters. The chapters are organised as

follows:

Chapter One deals with the introduction of the study. It includes the background to the study, statement of the problem, objectives of the study and the research questions. It also considers the significance of the study, delimitations and limitations of the study and definition of terms. Chapter Two covers review of related literature to the study. This includes

theoretical literature on succession planning, concept of succession planning and challenges of succession planning.

Chapter Three focuses on methodology of the study. It covers research design, population, sample and sampling techniques used for the study. It also describes the instrument used, pilot-testing as well as the procedure for the collection of the data and analysis of the data.

Chapter Four presents data analysis and discussion of research findings.

Chapter Five covers the summary of findings, conclusion drawn from the findings, recommendation made and suggestions for further research.



CHAPTER TWO

REVIEW OF RELATED LITEATURE

2.1. INTRODUCTION

This chapter aimed at providing a focus for the study as well as the basis for the assessment of the findings. The literature was reviewed on both theoretical and the empirical perspectives on succession planning in small and medium enterprises. Further, the literature review covers the historical perspectives of small and medium enterprises in Ghana, concept of succession planning, factors influencing succession planning in small and medium enterprises and challenges of succession planning in small and medium enterprises.

2.2. Concept of Small and Medium Scale Enterprise (SME)

The concept of SME has enjoyed some measure of controversy in the literature attempt at consistency and ambiguity. Consequently, there are as many definitions of SME as there are scholars and institutions defining it. Scholars are to use parameters such as number of employees, sales volume or turnover, capital small, medium or big scale. At one point some people, rightly contended that these parameters do not provide uniform benchmark for all industries and for all time. For instance, what

could pass for a big capital in food and beverage industry will certainly be small when compared to that in steel or aviation industry.

Further, what is a big capital today may become small in five years to come (Ezeife, 1998). Ezeh(1999), Nzelibe (1996) and Tinubu (2002) posited that SMEs in Nigeria consist of all business organizations whose total capital employed is above N1.5 million, but not more than N200 million including the cost of land and workforce capacity of between 11-300. Available statistics show that 95% of establishments in the organised manufacturing sector fall into SME category. NESG (2002) defined as based on the nature of business and magnitude. For instance, roadside artisans, petty traders, bottled water producers, bakers, local fabricators should constitute part of SMEs. In 1990 budget, the Federal Government of Nigeria defined small scale enterprises as those enterprises with annual turnover not exceeding N500, 000 or as those enterprises with capital investment not exceeding N 2 million. According to Umar (1997), the concept of the small size firm is a relative one and it depends mainly on both the geographical location and the nature of economy activity being performed. Umar asserted that because of flexibility nature of small scale enterprises, they are able to withstand adverse economic role since they are more of labour intensive. Therefore, they have lower capital costs that are associated with job creating and they are sine qua non for employment generation, encouragement for the use of local raw materials, employment generation and alleviation of poverty. According to Dasanayaka and Sardana (2007), SMEs engage in wide range of business activities in

agriculture, mining, fishing, industry or manufacturing, construction, retails and whole sales and services in rural, urban and estate settings by servicing local and international markets. SMEs are active in most of the industry sub sectors such as agricultural inputs or outputs business in rural areas to food and beverages business in city up to more advanced light engineering sectors such as computer, chemical, machinery, apparel and construction business in local and foreign markets. Most of these SMEs in developing and developed countries are one-person show or are run by few family individuals, usually relatives, friends or business partners, who take most of the decisions. Usually no distinction between private and business assets, and subjective and personal factors play a large role in decision-making. The personal stakes of SMEs in both countries have in their businesses are much higher than those of corporate executives in their companies (Dasanayaka, 2008a). There are very many definitions available based on assets, employees, skills, capital, turnover or revenue in local and exports markets, sophistication, innovation, productivity and growth orientation. However, most of these definitions are made according to organizational needs and purpose of interests about SMEs. Financial institutions, public sector authorities, non-governmental organizations (NGOs), trade and industry chambers, international organizations, researchers, SMEs service providers and consultancy firms have their own definitions based on their own criteria selection (Dasanayaka, 2008b). Most of these organisations used various terms for SMEs such as Small and Medium Scale Industries (SMIs), Micro Enterprises (MEs), Rural Enterprises (REs), Small and Medium Scale

Activities (SMAs), Cottage and Small Scale Industry (CSSI), Informal Sector Activities (ISAs), Micro and Small Scale Activities (MSSA), among others. Generally, enterprise is defined as any business activity or entity that engaged in industry, agri-business and services whether single proprietorship, partnership or corporate venture. In addition, various banks, financial institutions, donor agencies, NGOs, industry related task forces, trade and industry chambers adopted their own definitions for SMEs for their convenience and objectivity of studying SMEs. Nevertheless, almost all these definitions adopted their main criteria as number of employees, capital employed or total assets and turnover. These criteria have its own limitations overtime and need changes with technology improvement, productivity increases and inflation (Bari et.al, 2003).

SMEs in Ghana and Malawi can be categorised into urban and rural enterprises. The former can be sub-divided into 'organised' and 'unorganised' enterprises. The organised ones tend to have paid employees with a registered office whereas the unorganised category is mainly made up of artisans who work in open spaces, temporary wooden structures, or at home and employ little or in some cases no salaried workers. They rely mostly on family members or apprentices. Rural enterprises are largely made up of family groups, individual artisans, women engaged in food production from local crops. The major activities within this sector include:- soap and detergents, fabrics, clothing and tailoring, textile and leather, village blacksmiths, tinsmiting, ceramics, timber and mining, bricks and cement, beverages,

food processing, bakeries, wood furniture, electronic assembly, agro processing, chemical based products and mechanics (Liedholm & Mead, 1987; Osei et al, 1993, World Bank, 1992; Gray, Cooley & Lutabingwa, 1997). It is interesting to note that small scale enterprises make better use of scarce resources than large scale enterprises. Research in Ghana and many other countries have shown that capital productivity is high in SMEs (Steel, 1977; Child 1971). The reason for this is that SMEs are labour intensive with very small amount of capital invested. Thus, they tend to witness high capital productivity which is an economically sound investment. Thus, it has been argued that promoting the SME sector in developing countries will create more employment opportunities, lead to a more equitable distribution of income and will ensure increased productivity with better technology (Steel & Webster, 1991). Countries do not use the same definition for classifying their SME sector. Nor does a universal definition appear to be necessary. The definitions in use depend on the purposes those definitions are required to serve and the policies which govern the SME sector. However, the three parameters generally applied by most countries, singly or in combination are capital investment on plant and machinery; number of workers employed; and volume of production or turnover of business. Small business has represented a predominant form in the world of business (Beckhard & Dyer, 1983; Stern, 1986). Recently, more criteria has turned up to define the small and medium business, such as ownership percentage, voting power, power over strategic direction, involvement of multiple generations, and active

management by family members. According to the degree of owner and management involvement in the additional criteria, Shanker and Astrachan (1996) defined SME business into three groups: broad, definitions. The broadest and most inclusive middle and narrow definition requires that the family has some degree of effective control over strategic directions and that the business is at least intended to remain in the boundary of family. The family is generally known to have an indirect effect on their business by voting through board membership. Literature tends to indicate that succession planning (SP) fosters business continuity and increases the probability of a successful succession (Morris et al. 1997; Sharma et al. 2003a). Firstly, SP can enhance the transfer of business knowledge which is of key importance for performance in today's knowledge economy (Ernst & Young, 2006). Secondly, planning can help to minimise future tax liabilities and enhance the financial stability and value of the firm (Battisti & Massey, 2008). Without a SP, SME incumbents may not be able to realise the value of their substantial capital which is tied up in the 2007; Wallace & Bellingham, 2008). business (Weil This is inconsistent with the very high priority for retirement funds generation espoused by SME owner respondents in the Battisti and Massey,s (2008) survey. A third aspect relates to the common desire of family firm owners to have family continuity in the firm (Sharma et al., 2001; Sharma et al., 2003b). In this case, a SP may serve to identify an heir from amongst seemingly reluctant candidates due to their different training (other professional careers), their gender (Martin, 2001; Ibrahim

et al., 2008) and different lifestyle (Wallace, 2008). It may also minimise conflicts among apparent heirs, heightened emotions and sensitive estate issues (Sharma et al., 2001; Motwani et al., 2006) which may contribute to the lower survival rate of family businesses compared to non-family firms (Ibrahim et al., 2004). Despite these benefits of SP, Battisti and Massey's (2008) results show that only forty-seven per cent of New Zealand SME owners had an exit plan, which was more likely to be unwritten and informal than formal. Both family and non-family businesses provided similar reasons for not having an exit plan, such as prematurity or a lack of time to develop one (Battisti & Massey, 2008). This is unfortunate as literature indicates that it might take up to ten years to plan for business succession (Morris et al. 1997; Malinen 2004). SP is a lengthy process as there is no standard model or rules to fit all types of businesses (Sambrook 2005; Ip & Jacobs, 2006). However, the most common SP options are generational succession (Verry, 2008), selling the business (Sharma et al., 2003a) and management buy-out (entails a negotiated purchase agreement) or management buy-in (the incumbent stays in the business, but his or her responsibilities are scaled down) (Napoli, 2006). Because of SME owners' inability to find time to develop a SP, the complexity of the process, and the expected positive outcomes, it is suggested that these owners may need (professional) advice to engage in business succession planning (Battisti & Massey 2008). Chaston and Baker's (1998) empirically-tested model provided some more specific guidance as to the meaning of the concept in this context. Their thirteen advisor characteristics are those which SME

owners perceive provide positive impact on the success of the business. These include the advisor's personal attributes, business skills (Mole, 2002), contact regularity, personal information sharing, client and advisor (Kelly, 2008), close problem-solving relationship (Yaniv, 2004; Mole, 2007), availability (Wallace & Bellingham 2008), competence (Mole, 2002), consistency, openness, trustworthiness (Seal & Vincent-Jones, 1997; Lewis et al., 2007; Mole, 2007), commitment to problem-solving and listening ability.

In case of the narrowest definition, usually more than one family member from multiple generations involves with management; and they are responsible for all managerial decisions. By this definition, a majority of large companies, which was initially categorized as non-family business, should be actually included in the category of family business. From the above perspective, family business is a type of firm that is directly owned and managed by a family; at the same time, it can be indirectly controlled by a family and taken over by one generation to another as well. Family businesses are described as having a unique culture (Dyer, 1986). It seems to develop more relations of trust than non-family organization do due to the more organic ties between the family and the firm (Fiegener et al., 1994). The common understanding of family business is that the key attributes derive from the overlap of the three dimensions of family, management, and ownership (Davis & Tagiuri, 1989). Nevertheless, it seems appropriate to see that the family business also integrates the distinct organizational characteristics of both family and business. The

most studies on the definition of family business so far have focused on distinguishing the family business from non-family business. Although the family business contains these two aspects of somewhat contradictory values (Whiteside & Herz-Brown, 1991), there is an argument that family business must be viewed as a single system and the whole as greater than the mere sum of its separate parts (Hollander and Elman, 1988; Whiteside & Herz-Brown, 1991). Based on the concept of organizational life cycle, Gersick et al. (1997) emphasized the uniqueness of family business through the intersection of business goals and family goals throughout the process of management. Contributions of Family Business are also essential to the concept of free enterprise system. They exemplify the economic principle of private property, the political principle of pluralism, and the social principle of family (Ward, 1997). The family-owned businesses are rapidly emerging as a dominant form of business enterprises in both developing developed economies and thus their influences, as well as their numbers, are expected to be increasing unprecedentedly (Ryan, 1995). The family business is normally characterized to make more consistent investment (Ward, 1983) because its owners tend to be motivated by intangible factors greater than tangible outputs expected such as profit and growth and are not so pressed to show the brilliant profit each quarter, which is different from the firms in professional management (O'Farrell & Hitchen, 1988). Regarding remains employment, family business aspect the most influential contributor concerning job creation in virtually every

country (Sharma, 1997; Tan & Fock, 2001; Ward, 2004; Dyer, 1986; Lank, 1995; Miller et al., 2003). They create more than two thirds of all jobs and tend to employ more female, young, old, and part-time workers (American Institute of Certified Public Accountants, 1984). In terms of innovation, they create two and half times as much innovation as large firms on a per capita employed basis (Small Business Administration, 1983).

2.3. SUCCESSION PLANNING

According to Charan, Drotter and Noel (2001), succession planning a process of identifying and developing internal people with the potential to fill key business leadership positions in the company. Succession planning increases the availability of experienced and capable employees that are prepared to assume these roles as they become available. They further postulated that effective succession or talent-pool management concerns itself with building a series of feeder groups up and down the entire leadership pipeline or progression. Corporate Leadership Council (1998) posited that the fundamental to the succession-management process is an underlying philosophy that the top talent in the corporation must manage for the greater good of the enterprise. The Council posited that a "talent mindset" must be part of the leadership culture for these practices to be effective. Further, the Council suggested that SMEs benefit from succession planning by retaining high degrees of engagement among the larger leadership team in the business. To the Council, clear objectives are critical to establishing effective succession planning.

These objectives tend to be core to many or most companies that have well-established practices:

- 1. Identify those with the potential to assume greater responsibility in the organization,
- 2. Provide critical development experiences to those that can move into key roles,
- 3. Engage the leadership in supporting the development of high-potential leaders,
- 4. Build a data base that can be used to make better staffing decisions for key jobs,
- 5. Improve employee commitment and retention,
- 6. Meet the career development expectations of existing employees, and
- 7. Counter the increasing difficulty and costs of recruiting employees externally.

Atwood (2007) stated that management succession planning in the small and medium enterprises requires focus on the four key departments present in any business of any size: administration and finance, operations and customer fulfillment, sales and marketing. According to Atwood, when people think of management succession planning, they are not talking about who will own shares or assets in the future; they are talking about who is going to do the dirty, thankless and countless tasks which make the family business an asset worth

preserving in the first place. The current state of ownership succession planning among family businesses is decidedly mixed. Two-thirds of family business owners report a good understanding of the amount of estate taxes due upon their deaths, while one in five have no estate planning at all. Further, one in three junior generation family business members have no knowledge of their senior generation's transfer plans. The ownership succession planning issues that seem to be the most common areas of contention or omission in family business succession planning are technical mistakes, planning in a vacuum, leaving the business to the surviving spouse and the challenge of treating children equitably.

According to Suh, Yun-Hee and Park (2008), succession planning prepares the company and the employee for future needs. Thus, matching that employee's talents to current needs and training them for future responsibilities creates a cycle of anticipated growth and goals. A good succession plan maps out which employees are ready for new leadership roles as they become available, and when one employee leaves or is promoted to the next level, another employee is already trained and ready to step in where they are needed. Many offices accomplish this on a small scale by job sharing and cross training for every position. To them, a succession plan goes beyond planning for the training of employees to assume more responsible roles, providing that training and assessing each employee's ability to step in when a position becomes available. Succession planning is not just important for the company and its current employees, but also for investors, customers, the community,

the employees and their families. The impact of a company going out of business will splash like waves on every life it touches. Who is going to manage the business when you no longer work the business? How will ownership be transferred? Will your business even carry on or will you sell it? They suggested that business succession planning seeks to manage these issues, setting up a smooth transition between the owner and the future owners of the business. With family businesses, succession planning can be especially complicated because of the relationships and emotions involved and because most people are not that comfortable discussing topics such as aging, death, and their financial affairs. They concluded that this is why more than 70 percent of family-owned businesses and small and medium enterprises do not survive the transition from founder to second generation. The absence of a succession plan is usually fatal for any small and medium enterprises. The nature of succession plans is examined in terms of small business system dominant is characterised by an impersonal selection process based on merits. SMEs constitute the majority of business entities across the world. These businesses contribute forty-five to seventy percent of gross domestic product and employment in many countries. In developing and developed economies, SMEs are becoming the dominant type of enterprise. A crucial challenge in all SMEs is succession. It has been estimated that the worldwide survival rate of small and medium businesses to the second generation is only thirty percent, while fewer than fourteen percent survive beyond the third generation. Researchers have argued that this poor survival rate is due to economic

factors. However, in some cases, SMEs fail because they do not plan for business continuation. Therefore, the odds are stacked against SMEs all over the world and the situation seems even worse within Takoradi Metropolis for a successful hand-off to the next generation. The lack of longevity of SMEs remains a primary problem.

In Saudi Arabia, only ten percent of small and medium businesses survive to the third generation. The business's primary function is to ensure care and support of its members. Therefore, the business's main goal is to satisfy family member needs and social status. The fundamental function of a business is to generate goods and services through effective and efficient organizational behaviour and structure. According to Ward (2005), stories of SMEs failure are quite common, and known to all. The deleterious effects of nepotism are one example of such received wisdom. The issue of succession within a SME acquires a greater significance within the context of Saudi Arabian (SA) economy and culture, where small and medium enterprises (SMEs), which are mainly owned by families, constitute about ninety-five percent of all Saudi companies. However, more than ninety percent of SA SMEs do not survive to the third generation, and this leads to a considerable amount of capital destruction. In many SA family businesses, relationship is a principal factor determining management succession. The family in general determines the institutional values of family firms. When key managers are from the same family, their traditions, values and both verbal and non-verbal communication are established from a common source. Family relationships among the

owners and key employees influence the business as a whole. The entire family is affected by what happens in the business, and vice versa. The business system is mostly impersonal with profit maximisation as the primary driving force, while the family system is mostly personal with maintaining harmony and good family relationships as fundamental goals. Striking a balance between the objectives of family and business systems is a key to the success of family businesses. Succession is one vital business process that reveals which system (family or business) is dominant. In family businesses where the family system is dominant, the succession planning and process is often replete with nepotism or custom. In these firms, who will succeed to the helm of power is typically based on who is the eldest among the siblings or family relatives. In contrast, family businesses wherein the business system is dominant choose among those most prepared for the job. The family system-business system overlap in family firms is determined to a large extent by the culture of the society within which the family firms operate. The findings have shown that most SMEs particularly in Nigeria die within their first five years of existence. It was revealed that smaller percentage goes into extinction between the sixth and tenth year while only about five to ten percent of young companies survive, thrive and grow to maturity. Many factors have been identified likely contributing factors to the premature death. Key among this include insufficient capital, inadequate market research, lack of succession plan, inability to separate business and family or personal finances, lack of business strategy, inability to distinguish between revenue

and profit (Basil 2005). Succession is a reciprocal interaction that occurs between the predecessor and successor throughout the transferring process of leadership (Churchill & Hatten, 1987; Handler, 1989) and a common organizational contest to ensure the survival of incumbent firms in operation. It is more than an event and a multi-staged development that continues over time (Churchill & Hatten, 1987; Handler, 1990; Longenecker & Schoen, 1978). The succession requires the indepth analysis from a holistic perspective of family, management and ownership systems in order to understand adequately the interests of the different stakeholders (Brochhaus, 2004). It has been debated that three most significantly confronting themes regarding any small and medium business are succession. The continuous existence of small and medium business is the ultimate managerial challenge for owners, successors, family members and other stakeholders (Miller et al., 2003; Anroff et al., 2002). Succession imposes a wide variety of considerable changes over the present SME business: the current family relationship should be rectified; the traditional managerial initiative should be redistributed; and the enduring structure of governance must give way to a new form. This can lead to a conclusion that succession affects the founder, the successor, the family, the managers, the owners, the stakeholders via inevitable development of new family relationship, redistribution of influence patterns, management and ownership structure (Lansberg, 1988). Since the significance of successful transfer of business leadership is so considerable, approximately one third of the SME business literatures have identically dealt with the issues of succession (Sharma et al., 1996; Sharma et al., 2000). Success in business succession has two important implications such as the performance of the firm and the satisfaction of its stakeholders relevant to the context of succession (Sharma et al., 2003). A successful succession can sustain tacit knowledge that family members have inherently accumulated and maintain comparative advantages over non-family firms (Sharma et al., 2003). However, it is reported that the business performances after succession have not been always satisfactory (Birley, 1986; Ward, 1987; Davis & Harveston, 1998). Miller et al. (2003, p. 514) mention that "many intergenerational successions in the USA failed soon after the second generation took control". The failure in succession tends to derive from vague succession plan, incompetent or unprepared successors, and discords among family members.

Mattew et al. (1999) argues that the failure in succession is one of the key causes that many family firms fail to perpetuate and ensure family control. This implies the significance as well as the complexity in the practice of business succession. In order to ensure organisational continuance, many studies (Ward, 1987; Nam, 2002, 2005) highlighted the importance of succession planning. For the the sustainability purpose of family firms, their successful of succession of leadership proves essential. Accordingly, strategic preparation of succession planning turns out to be a crucial institutional foundation for the practice of succession. Furthermore, the effectiveness of succession closely concerns the sustaining of firm competitiveness and satisfactions of the stakeholders. It has been estimated the international perspective, merely that. in family business survive in the second generation, while fewer than 14% make it beyond the third generation (Flemming, 1997; Matthew et al., 1999). One of the main reasons for the high rate of failure comes from their managerial incapacity to cope with the complicated (and unavoidably emotional) procedure of passing over ownership as well as leadership (Matthew et al. 1999). Succession, which is defined as "the transfer of leadership from one family member to another" (Sharma et al., 2003, p.669), can be understood as a whole process that an owner officially retires from her/his business (Churchill & Hatten, 1987; Barnes & Hershon, 1976). Broadly, the succession is described as the interactions that take place between the predecessors and successors during the transition process (Churchill & Hatten, 1987; Handler, 1989; Logenecker & Schoen, 1978). Santiago (2000) found that succession in family businesses in the Philippines takes a different approach in comparison with that in Western countries. However, it seems that there is still a long way to go to crystallize and integrate the findings of the previous studies. Lack of a clearly or wellestablished theory in guiding research seems to be due to inherent problems in studies of small and medium and family firms, researchers' prescriptive propensity and difficulties of generalization of findings across studies (Smyrnios et al., 2003). According to the theory of planned behavior (Ajzen, 1991), the stronger the intentions to be engaged in a behaviour, the more likely should be the targeted behavior. In turn, the intention is postulated to be predicted by such three conceptually independent determinants as attitude toward behavior. subjective norm and perceived behavioral control. Both personal attitude toward outcomes of the behavior and perceived social norms are associated with the perceived desirability of performing the behavior. The perceived behavioral control reflects the degree of subject's perception that she/he can control the planned behavior, the perceived feasibility of performing behavior or subjects' perceptions of personal situational competence (Krueger & Carsrud, 1993). antecedents of the intention can be applied in of succession in SME business. In a understanding process sense, most initiators in a succession process are the owners or the incumbents. When an owner intends to perform succession, she or he perceives that a form of family firm through succession is desirable. Additionally, the owner, other family members, and other stakeholder groups make a degree of consensus that SME firms should remain competitive to sustain; it is related to the importance of interaction among all family members in the succession process (Sharma et al., 2003, p.670). Finally, the owner perceives that he or she is able to implement the succession process and hold controlling power that can influence the process. In this regard, Sharma et al. (2003) also assume that the owner possesses sufficient controlling power over the succession process as most families and the businesses have a majority of ownership of the firm. According to Sharma et al. (2003), succession planning of small and medium enterprises can occur in the following They identified the following as the ingredients context.

successful transition of succession planning of small and medium enterprises.

Role clarification of predecessor. The sound relationship between the incumbent and the successor is central the to success intergenerational transition. The perspectives on the role of outgoing owner managers vary. Some authors assert that the definitive withdrawal from the firm or very limited role during and after the successor's appointment (Churchill & Hatten, 1987; Gersick et al. 1999; Handler, 1989; Holland & Boulton, 1984; Holland & Oliver, 1992; Longenecker & Schoen, 2002) but others argue that the outgoing owner manager should be given some new pertinent roles (Malinen, 2001; Cabera-Suarez, 200 Chung & Yuen, 2003; Feltham et al., 2005; Cadieux, 2007). As an advisor, the retired founder can transfer his or her tacit knowledge and network of contacts to their successors (Malinen, 2001). According to process of transition, the expected role from the predecessor might be a supervisory role during joint-reign phase and consultant role during withdrawal phase (Handler, 1989). Even a founder predecessor with an entrepreneurial profile could become a new mentor if he or she personally is involved in the process of training and integrating the successor (Cadieux, 2007). The relations with the form can be maintained by overseeing firm's strategic decisions (Cabera-Suarez, 2005; Chung & Yuen, 2003; Feltham et al., 2005), making contacts with organizations as honorary positions, joining the board of directors, and acting as a firm's ombudsman (Cadieux, 2007). Thus, it is important to help the predecessor become

more self-aware to seek another role out of the firm (Hall, 1986). This is particularly critical to family firms, in that the incumbent's resistance to give up the status and power may be one of the severe barriers in accomplishing a smooth transition between one generation and the next. Because the founder fears losing control and is, concerned that retiring from the firm will mean a demotion in his or her role even within the family (Lansberg, 1988).

A successor must be regarded as one of Succession Autonomy. the most important stakeholders who influence the results of succession. The argument is that if a designated successor is reluctant to take over the family business, the succession is less likely to be successful (Venter et al., 2005). When a current owner-manager feels doubt about the ability and credibility of a successor, not only is the predecessor not likely to hand over completely, but wants to continue to take control of firms' decision processes even after the succession. Therefore, it may be worthwhile that a successor needs to be consulted with a former owner manager before the initial succession decision. If a potential successor does not find the discretions in management after the succession, the willingness of the potential successor to take over the family business may be less. In case of that the succession process has no other choices but to select a certain person as the successor, the succession processes may be worse-than-expected or even fail. Handler (1989) was the first to undertake a systematic and comprehensive study of the range of influences associated with next generation family members. Handler argues that a successful succession depends

on leaving the successor to make his or her own decisions. Goldberg and Woolridge (1992) suggest that the successor must take control of succession process. Shepherd and Zacharskis (2000) stress the successor's decision making processes. Interestingly, a study of successfully intergenerational completed transfer in Canada shows that successors are willingly asked to take over, contrary to other cases in general (Cadieux, 2007). The quality of relationship between the outgoing leader and successor is of great importance to the success of succession process (Brockhaus, 2004; Chrisman et al., 1998; Dickson, 2000; Lansberg & Astrachan, 1994). The high quality relationship is characterized by a high level of trust, mutual support, open and earnest communication, and acknowledgement of each other's achievement (Harvey & Evans, 1993; Neubaur & Lank, 1998).

Succession Communication. Studies of failed succession show that dispute among family members is frequently due to a lack of ground rules of selection. Minimal criteria relating to selection of prospective successor must be set in advance and acknowledged to all the concerned. The rules of the game and processes for the succession must be communicated early and clearly (Ambrose, 1983; Ward, 1997). In this it should be clearly communicated to all sense, concerned for the participants in the succession task force, selection criteria, procedures for continual evaluation and selection, time frame, and timing of succession and population of candidates (Breton-Miller et al., 2004). As mobilizing the succession planning process means to deal with the ambivalence of different stakeholders

(Lansberg, 1988), lack of both communication and trust may confound the whole process of succession (Handler, 1994). Thus, it may be assumed that the high quality of succession occurs when mutual respect, understanding, and communication between generations exist in the process of succession planning so that the family unit can feel an informed decision was made (Harvey & Evans, 1995).

Succession Justice. For the family business facing succession issues, justice relating to transition process should be taken importantly relationships among stakeholders account to maintain the including family members other than the predecessor and the 1990). The succession justice derives from successor (Ayres, recognition of stakeholders in terms of ability and fairness in selection process. The new successor in the family business is necessary to be confirmed by a majority of stakeholders, including key employees, investors, bankers, suppliers, and distributors (Harvey & Evans, 1995). If the successor sets in the new leadership position but is reluctantly or even not recognized as an appropriate person by the stakeholders, the successor are less likely to effectively take control of with little authority. Given this importance and business characteristics of the succession justice, thus, firms that prepare business succession should pay a great deal of attention to it. Taking account of the variables for succession context mentioned above, we develop a hypothesized model that is focused on the influences of the management practices and family characteristics on the succession context.

2.4. CONCLUSION

It is becoming increasingly apparent to government and policy makers that the role of small and medium scale enterprises is very important to the development and growth of any given economy. Small and medium scale enterprises will ensure efficient use of resources, employment creation, mobilisation of domestic saving and investment. However, one major problem facing SMEs is succession of the enterprises to the next generation.



CHAPTER THREE

METHODOLOGY

3.1. INTRODUCTION

This chapter describes the research design, the population, the sample and sampling procedure. It also describes the research instruments, pilot testing, the procedure for data collection and methods that are used for the data analysis.

3.2. RESEARCH DESIGN

Research design is a blue print which indicates how data relating to a given problem should be collected and analysed (Fraenkel & Wallen, 2000). The researcher employed the descriptive survey method. A descriptive research intends to present facts concerning the nature and status of a situation, as it exists at the time of the study. It is also concerned with relationships and practices that exist, beliefs and processes that are ongoing, effects that are being felt, or trends that are developing. In addition, such approach tries to describe present conditions, events or systems based on the impressions or reactions of the respondents of the research (Creswell, 2009). Gay (2005) states that descriptive survey involves the collection of data in order to answer questions concerning the current status of the problem. The major techniques or tools used in collecting data in this type of research are the questionnaire, interview and observation. According to Leedy (2006), descriptive

method of research looks with intense accuracy at the phenomena of the moment and then describes precisely what the researcher sees. To him, whatever the researcher observes at any one time is normal and under the same condition could be observed again in the future. The major techniques or tools used in collecting data in this type of research are the questionnaire, interview and observation. Since this study intends to find out succession planning of Small and Medium Enterprises in Takoradi Metropolis, the survey study is found most appropriate.

3.3. POPULATION

The target population included all the small and medium scale enterprises (SMEs) in Takoradi Metropolis. The population of the study consists of owners, family members, partners, management, spouses, relatives and workers of commerce or retail business, construction or manufacturing business and service business. At the time of study, there were 1250 Registered SMEs in the Takoradi Metropolis. These SMEs were made up of 252 Service enterprises, 830 Retail enterprises and 168 Construction enterprises.

3.4. SAMPLE AND SAMPLING PROCEDURE

The researcher used purposive, stratified, quota and convenience sampling techniques to select the respondents for the study. Purposive sampling technique is used by researchers to choose samples that are likely to be knowledgeable and informative about the phenomena under study (Babbie, 2008). 125 Small and Medium Enterprises that had 5 or more workers were purposively selected for the study. Stratified sampling is

when the population is divided into subgroups. In view of this, the researcher divided the population into owners and workers. Quota sampling is a type of non-probability sampling which involves a specific number of respondents being selected from the population. For this study the researcher selected 75 owners of SMEs representing 60% of the respondents for the study and 50 workers representing 40% of the respondents. The owners of the SMEs were given a higher percentage because succession planning directly involves them. A convenience sampling technique was used to select the 125 respondents of the SMEs. The decision to use a convenience sample instead of a probability sample is driven by cost.

3.5. SOURCES OF DATA

Data consist of primary and secondary were employed by the researcher. Primary data is by means of survey and structured questionnaire. The questionnaire was designed for respondents who are owners or proprietors and staff of SMEs in the Takoradi Metropolis. Secondary data was used in this study consisted of existing literature on succession planning of SMEs. To obtain these data, the researcher visited Ministry of Trade and Industry and Ghana Revenue Authority during the time of the study.

3.6. RESEARCH INSTRUMENTS

The sample of 125 SMEs in Takoradi Metropolis was used for the study. The study made use of quantitative data which was gathered through a personal survey of Small and Medium-scale Enterprises (SMEs)

at the 5 to 10 employee level, based on the information available from the Ghana Revenue Authority. A long list of SMEs was prepared to serve as a base from which 125 respondents non-randomly drawn for the study. Questionnaires were then personally giving to the selected respondents. The 125 sampled respondents filled-out questionnaires given to them. Respondents were guaranteed anonymity and it was believed that the questions were discreet. The questionnaire was developed from the literature review. Section 'A' of the questionnaire indicated biographic data of respondents. The section 'B' dealt with factors influencing succession planning of Small and Medium Enterprise and Section C looked at the strategies adopted to enhance succession planning in Takoradi Metropolis. Further, the questionnaire included open-ended and closed – ended items.

3.7. PILOT -TESTING

The pilot- test was done in line with Borg and Gall (1989) idea that there is the need to subject a new research instrument to field-testing with a population similar to that from which the sample for the study were taken. The instrument was pilot –tested at the Sekondi Metropolis. Sekondi was selected because of the similarity in its character to that of the Takoradi. Cronbach alpha was used to analyse the pilot test. The reliability coefficient of 0.87 was obtained. The pilot-test enabled the researcher to reshape, delete items that were ambiguous and reframe items that were unclear to respondents.

3.8. DATA COLLECTION PROCEDURE

An introductory letter was collected from my department head of Graduate School of Business Administration Studies enabled me to have a good rapport with the operators and owners of Small and medium Enterprises in Takoradi Metropolis. The purpose of the study was explained to the workers and this paved the way for the researcher to retrieve all the questionnaires from the respondent without difficulty. The instructions for completing the questionnaire explicitly appeared on the instrument, therefore no further instructions were needed when distributing the questionnaire. The researcher also availed herself to the respondents to answer questions that bordered on the study.

3.9. DATA ANALYSIS

The responses to the item on the questionnaires were analyzed using frequencies and percentages, with the use of Statistical Package for Social Science (SPSS) Version 16.0. To ensure consistency, the responses in the questionnaires were edited and coded. The responses for the open-ended questions were grouped based on common ideas that the respondents expressed. The results were presented using frequencies and percentages.

CHAPTER FOUR

RESULTS AND DISCUSSION

4.1. INTRODUCTION

This chapter presents analysis and discussion of the field data on succession planning of Small and Medium Enterprises in Takoradi Metropolis. The chapter is organised into three sections. The first section deals with the demographic data of the respondents. This includes the respondents' educational qualifications, the number of years respondents have worked with the SMEs. The second section, 'B', covers factors affecting succession planning of SMEs in Takoradi and the third section, 'C', is concerned with how succession planning in Small and Medium Enterprises can be enhanced.

4.2. DEMOGRAPHIC DATA OF RESPONDENTS

The researcher considered it appropriate to find out the demographic data of the respondents so as to give a clear focus of the study by providing relevant data concerning succession planning of small and medium enterprises in Takoradi Metropolis.

4.3. EDUCATIONAL QUALIFICATIONS OF THE RESPONDENTS

Data on the distribution of respondents' educational qualifications are showed in Table 4.1.

TABLE 4.1: The Distribution Of Respondents By Educational Qualifications

Educational qualification	No	%
Primary School	9	7.2
Junior Secondary School	13	10.4
Secondary School	43	34.4
Tertiary	60	48.0
Total	125	100

Source: Field survey, 2011

From Table 4.1, the distribution of the respondents by educational qualifications showed that 7.2% representing nine respondents out of 125 had received primary education. Thirteen percent of the 125 sampled respondents had Junior Secondary education whilst 34.4% of the respondents representing 43 had obtained Secondary school education. It is clear from Table 4.1 that 48.0 % representing 60 out of 125 respondents had tertiary education qualifications. This implies that majority of the respondents selected for the study had received education at various levels and therefore they could read and contribute to the discussion of the study.

4.4. NUMBER OF YEARS RESPONDENTS HAVE WORKED IN THE SMALL AND MEDIUM ENTERPRISES

Table 4.2: The Number Respondents Have Worked In The Smes

Response	No	%
2- 10 years	44	35.2
11 - 19 years	53	42.4
20 – 28 years	25	20.0
29 years and above	3	2.4
Total	125	100

Source: Field survey, 2011

As shown in Table 4.2, 44 representing 35.2% of the respondents had worked between 2 - 10 years in the small and medium enterprises, whereas 53 representing 42.4% of the respondents have worked for between 11- 19 years. Twenty- five representing 20.0% of the respondents had worked for 20 -28 years whilst three out of 125 respondents representing 2.4 % had worked for 29 years and above. Responses from the respondents on the numbers of years they have worked in the SMEs indicate that the workers are familiar with the phenomenon under study.

4.5. RESEARCH QUESTION 1: What Factors Affect Succession Planning In Small And Medium Enterprises In Takoradi Metropolis?Respondents' views on factors that influenced succession planning are presented in Table 4.3.

 Table 4.3:
 Factors Affecting Succession Planning In SMEs

Factors influencing	Agree (%)	Not s	sure (%) Dis	sagree(%)
succession planning				
Succession planning is based on goal of the SMEs	the 80(64	4.0)	27(21.6)	18(14.4)
Harmony in the SMEs relates to succession planning	98(7	78.4)	27(21.6)	0(0)
Proprietor's role after retirement to relate to succession planning	t 67(53	3.6)	33(26.4)	25(20.0)
Management practices relate to succession planning	45(3)	6.0)	10(8.0)	70(56.0)
SMEs do not institute succession planning due to lack of trust	on 108(8	86.4)	17(13.7)	0(0)
Succession planning increases the availability of experienced work		2.8)	15(12.0)	32(25.6)
SMEs provide training for the prospective successors	46(3	36.8)	10(8.0)	69(55.2)
Succession in SMEs is based on merits	28(2	22.4)	30(24.0)	67(53.6)
SMEs do not involve their famil in the operation of the business	ies 49(3	39.2)	18(14.4)	58(46.4)

Source: Field survey, 2011

Table 4.3 reveals that 80 out of the 125 respondents representing 64.0% agreed to the view that "Succession planning is based on the goal of the SMEs". Twenty-seven out of 125 respondents representing 21.6% were not sure that succession planning is based on the goal of the SMEs whereas 18 respondents representing 14.4% disagreed to the statement. It is evident from the analysis that the respondents agreed to the view that succession planning is based on the goals or clear objectives of the SMEs.

According to Charan, Drotter and Noel (2001), succession planning is a process of identifying and developing internal people with the potential to fill key business leadership positions in the company. Succession planning increases the availability of experienced and capable employees that are prepared to assume these roles as they become available. Corporate Leadership Council (1998) posited that the fundamental to the successionmanagement process is an underlying philosophy that the top talent in the corporation must manage for the greater good of the enterprise. To the Council, clear objectives are critical to establishing effective succession planning. These objectives tend to be core to many or most companies that have well-established practices: identify those with the potential to assume greater responsibility in the organization, provide critical development experiences to those that can move into key roles, engage the leadership in supporting the development of high-potential leaders, build a data base that can be used to make better staffing decisions for key jobs, improve employee commitment and retention, meet the career development expectations of existing employees, and counter the increasing difficulty and costs of recruiting employees externally.

The statement "Harmony in SMEs relates to succession planning" might be one of the factors that could influence succession planning at the SMEs. Thus, succession planning of SMEs depends on proprietor's relationship with his or her family and the employees. From Table 4.3, 98 out of 125 respondents representing 78.4% agreed with the statement that harmony in SMEs could relate to succession planning whereas 27 respondents representing 21.6% were not sure of the statement. The results showed that

none of the respondents representing 0.0% disagreed to the statement. The majority of the respondents agreed with the statement that when there is harmony in the SMEs, it might influence succession planning. According to Ward (2005), in many South Arabia family businesses, relationship is a principal factor determining management succession. The family in general determines the institutional values of family firms and when key managers are from the same family, their traditions, values and both verbal and non-verbal communications are established from a common source. Family relationships among the owners and key employees influence the business as a whole. The entire family is affected by what happens in the business, and vice versa. Atwood (2007) stated that two-thirds of family business owners report a good understanding of the amount of estate taxes due upon their deaths, while one in five have no estate planning at all. These statements imply that good relationship among members in the small and medium enterprises could be one of the factors affecting succession planning in SMEs.

As shown in Table 4.3, the statement "Proprietor's role after retirement relates to succession planning" revealed that 67 out of 125 respondents representing 53.6% agreed with the statement that proprietor's role after retirement could influence succession planning whereas 25 respondents representing 20.0% disagreed with the statement. However, 33 respondents representing 26.0% remained not sure to the statement that proprietor's role after retirement might affect succession planning in the SMEs. Thus, succession is a reciprocal interaction that occurs between the predecessor and successor throughout the transferring process of

leadership and a common organisational contest to ensure the survival of incumbent firms in operation (Churchill & Hatten, 1987; Handler1989). According to Brochhaus (2004), succession requires the in-depth analysis from a holistic perspective of family, management and ownership systems in order to understand adequately the interests of the different stakeholders. The quality of relationship between the outgoing leader and successor is of great importance to the success of succession process (Brockhaus, 2004; Chrisman et al., 1998; Dickson, 2000; Lansberg & Astrachan, 1994). The high quality relationship is characterised by a high level of trust, mutual support, open and earnest communication, and acknowledgement of each other's achievement (Harvey & Evans, 1993; Neubaur & Lank, 1998).

When asked if "Management practices relate to succession planning" in Small and Medium Enterprises, 45 out of 125 respondents representing 36.0% agreed that management practices could result in succession planning whereas 10 respondents representing 8.0% were not sure of the statement and 70 respondents representing 56.0% disagreed to this assertion. The majority of the respondents disagreed that management practices might influence succession planning in SMEs. Thus, succession in SMEs is influenced by owner's preference and choice or sometimes family's decision on who to succeed.

The statement "SMEs do not institute succession planning due to lack of trust" was one of the factors that could influence succession planning in SMEs. The result showed that 108 out of 125 respondents representing 86.4% agreed to the statement but 10 respondents representing 8.0 % were

not sure about the statement that management do not institute succession planning due to lack of trust. Further, seven respondents representing 5.6% disagreed with the statement "SMEs do not institute succession planning due to lack of trust". Thus, majority of the respondents agreed that "SMEs do not institute succession planning due to lack of trust" was one of the factors that could affect succession planning. In the light of this, Domfeh (2011) posited that small and medium scale businesses in Kumasi need succession plan to sustain their operations. Further, retirement seems not to be part of their business plan, let alone plans to hand over the business. Studies suggest that establishing a sound business succession plan is beneficial for business owners. Domfeh (2011) observed that lack of trust is a major factor influencing SME owners not to institute succession plans. Statistics show that about 70 percent of SME businesses collapse because of lack of succession planning. In support of the above statement, Harvey and Evans (1993), Neubaur and Lank (1998) stated that the high quality relationship is characterised by a high level of trust, mutual support, open and earnest communication and acknowledgement of each other's achievement is a key factor in succession planning in any small and medium enterprises.

When asked if "SMEs provide training for prospective successor", 46 out of 125 respondents representing 36.8% agreed whereas 10 respondents representing 8.0% were not sure of the statement. The Table 4.3 showed that 69 respondents representing 55.2% disagreed to this assertion that "SMEs provide training for prospective successor". The majority of the respondents disagreed that the SMEs do not provide training for

prospective successor. According to Mamman and Oluyemi (1994), managers and directors of some SMEs often lacked the necessary expertise and experience. Further, they stated that recruiting good staff was often difficult for the SMEs because the established SMEs could usually offer the most talented SMEs officials better career prospects. This situation rose because many SMEs fill managerial positions with their families and therefore they believed that it was not necessary to spend on training people for the position.

From the table, the statement "succession in SMEs is based on merits" was identified as one of the factors influencing succession planning in the SMEs. The findings revealed that 28 out of 125 respondents representing 22.4% agreed with the statement succession in SMEs is based on merits could affect succession planning in the SMEs. The respondents selected for the study agreed that inadequate SMEs managers might be the cause of succession planning. According to Mamman and Oluyemi (1994), managers and directors of some SMEs often lacked the necessary expertise and experience. However, 14 respondents representing 18.7 % were not sure if inadequate qualified SMEs could influence succession planning at the SMEs and three of the respondents representing 4% disagreed with statement.

The table further revealed that 49 respondents representing 39.2 % agreed with the statement that SMEs do not involve their families in the operation of the business could affect of succession planning at the SMEs whereas 18 respondents representing 14.4 % were not sure with the statement. The findings showed that 58 respondents representing 58% disagreed with the

statement that SMEs' do not involve their families in the operation of the business could affect of succession planning. It is clear from the analysis that the majority of the respondents disagreed that SMEs' do not involve their families in the operation of the business could affect succession planning.

4.6. RESEARCH QUESTION 2: What Factors Can Be Adopted To Enhance Succession Planning In The Small And Medium Enterprises?

Respondents' views were sought on strategies that can enhance succession planning of small and medium enterprises are provided in Table 4.4.

Table 4.4: Strategies that can be adopted to enhance succession planning in small and medium enterprises

Strategies to enhance succession planning	No	%
Government can make laws to regulate succession planning in SMEs	43	34.4
Training of prospective successors for future needs	87	69.6
Succession document can be put in place upon founder' retirement	70	56.0
Succession in SMEs should be based on merits	68	54.4

Source: Field survey, 2011

Table 4.4 revealed that 43 out of 125 respondents representing 34.4% of the sampled respondents responded that Government can formulate laws to regulate succession planning of SMEs. This implies that the SMEs could float shares as one of the ways of overcoming its succession planning.

According to Table 4.4, 87 out of the 125 respondents representing 69.6% responded that the Training on leadership and management can be organised for prospective successors of the business.

Table 4.4 again reveals that 70 out of the 125 respondents representing 56.0% of the sampled respondents responded that SMEs owners can prepare the business for smooth transfer upon their retirement through proper management practices by documenting it.

The findings in Table 4.4 depicted that 68 respondents representing 54.4% of the sampled workers suggested that Succession planning in SMEs should be characterized by an impersonal selection process based on merits. According to Suh, Yun-Hee and Park (2008), With family businesses, succession planning can be especially complicated because of the relationships and emotions involved and because most people are not that comfortable discussing topics such as aging, death, and their financial affairs. They concluded that this is why more than 70 percent of familyowned businesses and small and medium enterprises do not survive the transition from founder to second generation. The absence of a succession plan is usually fatal for any small and medium enterprises. The nature of succession plans is examined in terms of small business system dominant is characterised by an impersonal selection process based on merits. SMEs constitute the majority of business entities across the world. These businesses contribute forty-five to seventy percent of gross domestic product and employment in many countries. In developing and developed economies, SMEs are becoming the dominant type of enterprise. A crucial challenge in all SMEs is succession. It has been estimated that the

worldwide survival rate of small and medium businesses to the second generation is only thirty percent, while fewer than fourteen percent survive beyond the third generation. Researchers have argued that this poor survival rate is due to economic factors. However, in some cases, SMEs fail because they do not plan for business continuation.



CHAPTER FIVE

SUMMARY, CONCLUSIONS AND RECOMMENDATIONS

This chapter presents the summary of findings, conclusions drawn from findings, and recommendations and suggestions made for further research.

The study sought to investigate the succession planning in Small and Medium Enterprises. Two research questions were raised to guide the study. The study focused on areas such as factors influencing succession planning in small and medium enterprises and how succession planning can be enhanced in Takoradi Metropolis.

The study adopted the descriptive survey design. The accessible population comprised workers and proprietors of the SMEs. Purposive, stratified, quota and convenience sampling techniques were used to select respondents. A total of 125 respondents were used as sample for the study. The study employed the use of questionnaire for the study. The questionnaire was tested at Sekondi Metropolis. The cronbach alpha was used for the pilot-tested and the alpha coefficient obtained was 0.87. Data gathered were analysed and presented with the help of percentages and frequency tables.

5.1. SUMMARY OF FINDINGS

The principal findings of the study are as follows:

- The study showed that lack of trust was one the factors affecting succession planning in the SMEs. This situation arose because many SMEs do not have clear objectives of succession planning or do not provide to the successors with the right training needed to manage the business.
- 2. The study found out that succession was not based on merit or succession was based on the relationship with the founder of the business.
- 3. The study revealed that good management practices and clear objectives of the SMEs in terms of who would take over or assume position when retirement or death could relate to succession planning.
- 4. The study identified many SMEs do not involve their family or relative in the operations of the business.

5.2. CONCLUSION

From the findings of the study, the following conclusions are drawn.

- 1. It is clear from the study that SMEs needs succession planning to ensure continuity of the enterprise. The SMEs can document successors upon the retirement or death of the founders or owners.
- The study revealed that many SMEs do not institute succession planning due to lack of trust. This situation occurs because many SMEs do not train people for succession role.
- 3. Another conclusion that can be drawn from the findings is that succession should be based on merit and competence of the successor.

5.3. RECOMMENDATIONS

The following recommendations are proposed for the attention of the founders, successors and managers of Small and Medium Enterprises.

- 1. The study recommends that the SMEs should have succession laws to ensure smooth transition as well as the roles of the successor.
- 2. The study recommended that SMEs should train prospective successors for the enterprises.
- 3. Finally, the study recommends that successors for the business should be based on merit to ensure that right and experienced people manage the business.

5.4. SUGGESTION FOR FURTHER STUDY

SCM SCAPSURE

A global study on the effect of succession planning in the Small and Medium Enterprises in Ghana will help in the understanding of the vital role SMEs play in the Ghanaian economy.

REFERENCES

Arieu, G. S. (2000). Effectiveness of entreprenership development programs in the creation and growth of smes in Sri Lanka, M.SC. Research Paper (Unpublished), University of Sri Jayewardenepura, Colombo

Arieu. G. S. (2003). Small And Medium Enterprise Developmentin Sri Lanka: A Review 133

Ariwa, S.A.S (2005). The Business Entrepreneur: A Guide To Entrepreneurial Development. Kaduna: Scopy Publishing.

Atwood, I. S. (2007). Relationship influencers: Determination of affect in the provision of advisory services to SME Sector Firms. *Journal of European Industrial Training*, Vol. 22 No. 6, Pp. 249-567

Basil, A. N. O. (2005). Small and medium enterprises (SMEs) in Nigeria: Problems and prospects, PhD. Thesis

Battisti, M. & Massey, C (2008). Succession planning In NZ SMEs: An exploratory study in Small Enterprise Association Of Australia And New Zealand (SEAANZ) Conference, Sydney

Besson, D. & Haddadj, S. (2002). Dysfunctions in owner-manager succession process in family firms and how a seam intervener-researcher can address them. *Journal of Organizational Change*, Vol. 16 No. 1, Pp. 83-90

Birnberg, J.G., Shields, M. D. & Young, S. M. (1990). The case for multiple methods in empirical management accounting research, *Journal Of Management Accounting Research*, Vol. 2, Pp. 33-66

Borg, W.R. And Gall. M. D. (1989). *Educational research*. New York: Longman Brochhaus, D. (2004), 'Succession choice', *Succeed Magazine*, P. 18

Carsrud, A. L. (1994). Meanderings of a resurrected psychologist or lessons learned in creating a family business program", *Entrepreneurship Theory And Practice*, Vol. 19 No. 1, Pp. 39-48

Chaston, I. & Baker, S. (1998). Business succession considerations [online] http://www.northwesternmutual.com/financial-guidance/business/as-your-business-matures/succession-planning.aspx [accessed 20th June2011]

Churchill, N. C. & Lewis, V. L. (1983). The five stages of small business growth, Harvard Business Review

Corporate Leader Council (1998), "Problems in transfer of business experienced by finnish entrepreneurs", CLC, Vol. 11 No. 1, pp. 130-9

Creswell, J. W. (2003). Research design: qualitative, quantitative, and mixed methods approaches (2nd Ed.). Thousand Oaks, Ca: Sage Publications, Inc

Davis, B.D., Pawlowski, S. D., & Housten, A. (2006). Work commitments of baby boomers and gen-xers in the it profession: generational differences or myth?" *The Journal of Computer Information Systems*, Vol. 46 No. 3, Pp. 43-9

De Kok, J. And Uhlaner, L. (2001), "Organisation context and human resource management in the small firm", *Small Business Economics*, Vol. 17 No. 4, Pp. 273-91

Domfeh, J. A (2011). The fate of small and medium enterprises in Kumasi. Daily Graphic, May 22, 2011 p.16

Earle, H.A. (2003), "Building a workplace of choice: using the work environment to attract and retain top talent", *Journal of Faculties Management*, Vol. 2 No. 3, Pp. 244-57

Ernst Young (2006), 'The knowledge economy', Ernst Young, New Zealand.

Fransden, B.M. (2009), "Leading by recognizing generational differences", long-term living: For The Continuing Care Professional, Vol. 58 No. 2, Pp. 34-5.

Gascoigne, C. (2002), 'Relative advantages of a personal passion: family businesses', In Financial Times, P. 15

Half, R. (2001), "Accountants storm the boardrooms", *Chartered Accountants Journal*, Vol. 80 No. 9, P. 28

Harris, P. & Kirkwood, J. (2006), 'Pass it on', Dunedin City Council and University of Otago, Dunedin

Hayes, K. (2006), Succession planning invest in success, a guide for securing the future for your business, Brookers Ltd, Wellington

Haylock, J. (2004), "Succeed with succession", Chartered Accountants' *Journal Of New Zealand*, Vol. 83 No. 3, P. 62

Howe, N. & Strauss, W. (2000), Millen*nials rising- the next great generation*, Vintage House, New York

Ibrahim, N.A., Angelidis, J.P. & Parsa, F. (2008), "Strategic management of family businesses: current findings and directions for future research", *International Journal Of Management*, Vol. 25 No. 1, Pp. 95-110

Ip, B. & Jacobs, G. (2006), "Business succession planning: A review of the evidence", *Journal of Small Business and Enterprise Development*, Vol. 13 No. 3, Pp. 326-501

Kayanula, C. & Quartey, N. R. (2002). Perceptions of SME growth onstraints in

Ghana. Journal Of Small Business Management, Vol. 40 (1), Pp.58-65.

Kelly, J. (2008), "Effective succession planning for CAS in public practice", *Chartered Accountants' Journal Of New Zealand*, Vol. 87 No. 2, Pp. 22-33

Kim, H. & Devaney, A.A. (2003), "The expectation of partial retirement among family business owners", *Family Business Review*, Vol. 16 No. 3, Pp. 199-210

Kyeremanteng, C. (2007), Succession planning in SMEs, *International Small Business Journal*, Vol. 24 No. 5, Pp. 47-97

Leach, R., Phillipson, C., Biggs, S. & Money, A. (2008), "Sociological perspectives on the baby boomers", *Quality In Aging*, Vol. 9 No. 4, Pp. 19-26

Leedy, P. D.(1985). Practical research: planning and design (4th Ed.) New York:

Macmillan Publishing Company

Lewis, K., Massey, C., Ashby, M., Coetzer, A. & Harris, C. (2007), "Business assistance for SMEs: New Zealand owner-managers make their assessment", *Journal of Small Business And Enterprise Development*, Vol. 14 No. 4, Pp. 55-66

Liedholm, C & Mead, D. (1987). Small scale industries in Developing Countries: Sources of finance and importance, *Ilorin Jurnal of Business and Social Sciences*,

Litz, R. (1995), "The family business: Toward definitional clarity", *Family Business Review*, Vol. 8 No. 2, Pp. 71-81

Malinen, P. (2004), "Problems in transfer of business experienced by Finnish Entrepreneurs", *Journal of Small Business And Enterprise Development*, Vol. 11 No. 1, Pp. 130-137

Martin, L. (2001), "More jobs for the boys? Succession Planning in SMEs." Women In Management Review, Vol. 16 No. 5, Pp. 222-310

Mcintyre- Smith, S. (2005), 'Practising succession planning', CA Magazine, Pp. 31-2, 5

Mole, K.F. (2002), "Business advisers' impact on SMEs: An agency theory approach", *International Small Business Journal*, Vol. 20 No. 2, Pp. 137-157

Mole, K.F. (2007), "Tacit knowledge, heuristics, consistency and error signals", *Journal of Small Business And Enterprise Development*, Vol. 14 No. 4, Pp. 582-601

Morris, M.H., Williams, R.W., Allen, J.A. & Avila, R.A. (1997), "Correlates of success in family business transitions", *Journal of Business Venturing*, Vol. 12, Pp. 385-401

Motwani, J., Levenburg, N.M., Schwarz, T.V. & Blankson, C. (2006), "Succession planning in SMEs", *International Small Business Journal*, Vol. 24 No. 5, Pp. 471-495

Napoli, R. (2006), 'Selling out', CA Magazine, Pp. 43-48

Nigeria Economic Summit Group (2000). A survey of micro and medium enterprises, NESG Digest , Lagos

Owen, W. (2007), 'The X and Y of The Generation Game', The Independent, 21 November, P.8.

Perry, M., Lardow, A. & Massey, C. (2006), 'Telling tales: A report on NZ SMEs and their Relationships with banks and accountants', Massey University, Wellington.

Ramsden, M. & Bennett, R.J. (2005), "The benefits of external support to SMEs, "hard" versus "soft" outcomes and satisfaction levels", *Journal Of Small Business And Enterprise Development*, Vol. 12 No. 2, Pp. 227-243

Reeb, W.L. (2005), "Nothing succeeds like succession", *Journal of Accountancy*, Vol. 200 No. 1, Pp. 63-65,

Reilly, K., Kirkwood, J. & Wallace, A. (2008), 'A professional opinion', Succeed

Magazine, Pp. 12-40

Sambrook, S. (2005), "Exploring succession planning in small, growing firms", *Journal of Small Business And Enterprise Development*, Vol. 12 No. 4, Pp. 579-594.

Sharma, P., Chrisman, J.J. & Chua, J. (2003a), "Predictors of satisfaction with the succession process in family firms", *Journal Of Business Venturing*, Vol. 18, Pp. 667-687

Sharma, P., Chrisman, J.J. And Chua, J. (2003b), "Succession planning as planned behavior:some empirical results", *Family Business Review*, Vol. 16 No. 1, Pp. 1-15

Sharma, P., Chrisman, J.J., Pablo, A. & Chua, J. (2001), "Determinants of initial satisfaction with the succession process in family firms: a conceptual model", *Entrepreneurship Theory And Practice*, Vol. 25 No. 3, Pp. 17-35

Smyrnios, K.X., Romano, C.A. & Dana, L.E. (2000), "10 steps to succession planning", *Australian CPA*, Vol. 70 No. 11, Pp. 44-46.

Umar, A. S. (1997). Effectiveness of entreprenership development programs in the creation and growth of SMEs In Sri Lanka, Msc. Research Paper (Unpublished), University Of Sri Jayewardenepura, Colombo

Verry, D. (2008), 'Succession choice', Succeed Magazine, P. 18

Wallace, A. (2008), 'Tackle the future', Succeed Magazine, P. 16

Wallace, A. & Bellingham, M. (2008), 'Succession planning for your business.' in professional development course papers, New Zealand Institute Of Chartered Accountants, Wellington

Wang, Y., Watkins, D., Harris, N. & Spicer, K. (2004), "The relationship between succession issues and business performance. *Evidence From Uk Family SMEs*",

Whitney, D. (2005), 'Boomer or bust? here comes the wall of business.' *Succeed Magazine*, Pp. 4-7

Yun-Hee, S. & P (2008). Small And Medium Enterprise Development in Sri Lanka: A Review 133



APPENDIX A

Kwame Nkumah University Of Science And Techhnology Institute For Distance Learning

Succession Planning In Small And Medium Enterprises (Smes) In Takoradi Metropolis

Succession Planning Had Been Identified As A Vital Tool To Ensure Efficiency In Continuity In Business. The Questionnaire Has The Objective Of Seeking Information On Succession Planning Of Smes In Takoradi Metropolis. The Study May Help In Improving The Succession Of The Enterprise. You Are Therefore Invited To Participate In The Study By Filling In The Attached Questionnaire.

You Will Be Contributing Immensely Towards The Success Of The Study If You Kindly Answer The Following Questions As Frankly As You Can. Your Response Will Be Accorded The Needed Confidentiality.

Section A: Demographic Data

Please Tick [$\sqrt{\ }$] The Appropriate Box That Corresponds To The Your Response.
1. Gender: A. Male [] B. Female []
2. Educational Qualification: A. Primary/ JHS [] B. Secondary []
C. Tertiary [] D. Others
3. Working Experience: A. Below 5 Years [] B. 5-10 Years []
C. Above 10 Years []
Section B: Succession Planning
The Items Below Are Statements On Factors Affecting Succession Planning.
Please, Tick (√) The Statement That Best Describe Your View On The Succession
Planning.
Key:
A – Agree
NS – Not Sure
D – Disagree
Factors Affecting Succession Planning A NS D

Goal Of The Smes 5 Harmony In The Smes Relates To Succession Planning 6 Proprietor's Role After Retirement To Relate To Succession Planning 7 Management Practices Relate To Succession Planning 8 Smes Do Not Institute Succession Planning Due To Lack Of Trust 9 Succession Planning Increases The Availability Of Experienced Workers 10 Smes Provide Training For The Prospective Successors 11 Succession In Smes Is Based On Merits 12 Smes Do Not Involve Their Families In The Operation Of The Business	4	Succession Planning Is Based On The					
Planning 6 Proprietor's Role After Retirement To Relate To Succession Planning 7 Management Practices Relate To Succession Planning 8 Smes Do Not Institute Succession Planning Due To Lack Of Trust 9 Succession Planning Increases The Availability Of Experienced Workers 10 Smes Provide Training For The Prospective Successors 11 Succession In Smes Is Based On Merits 12 Smes Do Not Involve Their Families In The		Goal Of The Smes					
Planning 6 Proprietor's Role After Retirement To Relate To Succession Planning 7 Management Practices Relate To Succession Planning 8 Smes Do Not Institute Succession Planning Due To Lack Of Trust 9 Succession Planning Increases The Availability Of Experienced Workers 10 Smes Provide Training For The Prospective Successors 11 Succession In Smes Is Based On Merits 12 Smes Do Not Involve Their Families In The	5	Harmony In The Smes Relates To Succession				-	
6 Proprietor's Role After Retirement To Relate To Succession Planning 7 Management Practices Relate To Succession Planning 8 Smes Do Not Institute Succession Planning Due To Lack Of Trust 9 Succession Planning Increases The Availability Of Experienced Workers 10 Smes Provide Training For The Prospective Successors 11 Succession In Smes Is Based On Merits 12 Smes Do Not Involve Their Families In The							
To Succession Planning Management Practices Relate To Succession Planning Smes Do Not Institute Succession Planning Due To Lack Of Trust Succession Planning Increases The Availability Of Experienced Workers Smes Provide Training For The Prospective Successors Succession In Smes Is Based On Merits Smes Do Not Involve Their Families In The	6	-				-	
Planning 8 Smes Do Not Institute Succession Planning Due To Lack Of Trust 9 Succession Planning Increases The Availability Of Experienced Workers 10 Smes Provide Training For The Prospective Successors 11 Succession In Smes Is Based On Merits 12 Smes Do Not Involve Their Families In The							
Planning 8 Smes Do Not Institute Succession Planning Due To Lack Of Trust 9 Succession Planning Increases The Availability Of Experienced Workers 10 Smes Provide Training For The Prospective Successors 11 Succession In Smes Is Based On Merits 12 Smes Do Not Involve Their Families In The	7	Management Practices Relate To Succession		-			
Due To Lack Of Trust 9 Succession Planning Increases The Availability Of Experienced Workers 10 Smes Provide Training For The Prospective Successors 11 Succession In Smes Is Based On Merits 12 Smes Do Not Involve Their Families In The							
Due To Lack Of Trust 9 Succession Planning Increases The Availability Of Experienced Workers 10 Smes Provide Training For The Prospective Successors 11 Succession In Smes Is Based On Merits 12 Smes Do Not Involve Their Families In The	8	Smes Do Not Institute Succession Planning				-	
Availability Of Experienced Workers 10 Smes Provide Training For The Prospective Successors 11 Succession In Smes Is Based On Merits 12 Smes Do Not Involve Their Families In The							
10 Smes Provide Training For The Prospective Successors 11 Succession In Smes Is Based On Merits 12 Smes Do Not Involve Their Families In The	9	Succession Planning Increases The					
Successors 11 Succession In Smes Is Based On Merits 12 Smes Do Not Involve Their Families In The		Availability Of Experienced Workers	3				
11 Succession In Smes Is Based On Merits 12 Smes Do Not Involve Their Families In The	10	Smes Provide Training For The Prospective					
12 Smes Do Not Involve Their Families In The		Successors					
12 Smes Do Not Involve Their Families In The			1	-	3		
	11	Succession In Smes Is Based On Merits	13	\neq	7		
Operation Of The Business	12	Smes Do Not Involve Their Families In The	3	7		-	
		Operation Of The Business			\		
		7777		7	-		
					And Me	edium Eı	nto
13 3 3		C.					
13. How Can Succession Planning Be Enhanced In The Small And Medium Ent			10110	vide i	our bug	Sestions	
13. How Can Succession Planning Be Enhanced In The Small And Medium End (Smes)? (Please, Use Of The Space Provided Below To Provide Your Suggestions Of the Space Provided Below To Provide Your Suggestion Provided Below To Provide Your Suggestion Provided Provided Below To Provided Provide	10 1	mance succession Framing in Sines).					
13. How Can Succession Planning Be Enhanced In The Small And Medium Ent (Smes)? (Please, Use Of The Space Provided Below To Provide Your Suggestions Of the Space Provided Below To Provide Your Suggestion Provided Below To Provide Your Suggestion Provided Provided Below To Provide Your Suggestion Provided P							
13. How Can Succession Planning Be Enhanced In The Small And Medium End (Smes)? (Please, Use Of The Space Provided Below To Provide Your Suggestions Of the Space Provided Below To Provide Your Suggestion Provided Below To Provide Your Suggestion Provided Provided Below To Provided Provide			• • • • • • • • • • • • • • • • • • • •	• • • • • • • •		•••••	
Section C Strategies To Enhance Succession Planning In Smes 13. How Can Succession Planning Be Enhanced In The Small And Medium Ent (Smes)? (Please, Use Of The Space Provided Below To Provide Your Suggestions Companies of Enhance Succession Planning In Smes).	• • • • • • • • • • • • • • • • • • • •		•••••				••••
13. How Can Succession Planning Be Enhanced In The Small And Medium End (Smes)? (Please, Use Of The Space Provided Below To Provide Your Suggestions Of The Space Provided Below To Provide Your Suggestion Provided Below To Provide Your Suggestion Provided Provided Below To Provided Provide	•••••		•••••	••••••	••••••	•••••	••••
13. How Can Succession Planning Be Enhanced In The Small And Medium End (Smes)? (Please, Use Of The Space Provided Below To Provide Your Suggestions Of The Space Provided Below To Provide Your Suggestion Provided Below To Provide Your Suggestion Provided Provided Below To Provided Provide	•••••		•••••	•••••	•••••		••••
13. How Can Succession Planning Be Enhanced In The Small And Medium End (Smes)? (Please, Use Of The Space Provided Below To Provide Your Suggestions Of The Space Provided Below To Provide Your Suggestion Provided Below To Provide Your Suggestion Provided Provided Below To Provided Provide	•••••		•••••	• • • • • • • • • • • • • • • • • • • •	•••••		••••
13. How Can Succession Planning Be Enhanced In The Small And Medium End (Smes)? (Please, Use Of The Space Provided Below To Provide Your Suggestions Of The Space Provided Below To Provide Your Suggestion Provided Below To Provide Your Suggestion Provided Provided Below To Provided Provide				•••••	• • • • • • • • • • • • • • • • • • • •		

.....

.....

Thank You For Your Time!

